Author's response to reviews

Title: Differences in the perceived role of the healthcare provider in delivering Vascular Health Checks: a Q methodology study

Authors:

Stephanie A Honey (S.A.Honey@leeds.ac.uk)
Louise D Bryant (L.D.Bryant@leeds.ac.uk)
Jenni Y Murray (J.Y.Murray@leeds.ac.uk)
Kate KM Hill (K.M.Hill@leeds.ac.uk)
Allan AO House (A.O.House@leeds.ac.uk)

Version: 2 Date: 18 March 2013

Author's response to reviews:

Dear Sir

I have added information about the health professional literature review we carried out. This will enable readers to replicate our methods.

These changes appers on page 4, paragraphs 2 and 3

Kind regards

Dr Stephnaie Honey